Author’s response to reviews

Title: Design and baseline data in the BAriatic surgery SUbstitution and Nutrition study (BASUN): A 10 year prospective cohort study

Authors:

Gudrun Höskuldsdottir (gudrun.hoskuldsdottir@vgregion.se)
Karin Mossberg (karin.mossberg@vgregion.se)
Ville Wallenius (ville.wallenius@gastro.gu.se)
Angelos Al Nimer (angelos.al.nimer@vgregion.se)
Wiveka Björkvall (wiveka.bjorkvall@vgregion.se)
Sören Lundberg (soren.lundberg@gastro.gu.se)
Carl-Johan Behre (carljohan.behre@vgregion.se)
Malin Werling (malin.werling@gmail.com)
Björn Eliasson (bjorn.eliasson@medic.gu.se)
Lars Fandriks (lars.fandriks@gastro.gu.se)

Version: 2 Date: 06 Feb 2020

Author’s response to reviews:

Dear Editor,

Thank you for your time on reviewing our paper. We now submit a revised, clean version as well as responses to individual comments made.

We hope that these revisions to our paper are satisfactory.

Best regards,
Gudrun Höskuldsdottir, MD
Corresponding author.

1. The last section of the abstract should be Trial Registration:
Response: The trial registration number and date are included in the abstract.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript:
Response: We have added information on the funding bodies roll in the study to the funding section.

3. We notice that some authors are missing from the authors' contributions section:
Response: We have clarified the authors’ contributions in this sections

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text:
Response: Thank you, we now submit clean files without tracked changes.